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Minutes for the November 24, 2010, CERES  

Production Oversight Board Meeting  
 

 

Meeting began at 1:00 p.m. in Building1250 Room 116. 

Meeting Notes 

There were no comments or corrections on the minutes from the last meeting. 

Announcements  
None 

 

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting) 

TISA Averaging 795 – 795 has been delivered to CM, but will be redelivered early next week.  

No other comments; SCCR to be approved. 

 

TISA Gridding 816 – Previously delivered files for 2002 & 2003 need to be deleted from 

production.  They’ve already been deleted from the SSI&T area, and this SCCR has already 

been approved and released. 

 

TISA Gridding 817 – Terra & Aqua Edition1A SFC sample read package.  No comments; SCCR 

to be approved. 

 

GGEO 818 – Delta delivery.  Already approved and released. 

 

Systems 

No one available to discuss systems. 

Status of Open SCCRs 

716 - 2 more weeks to complete SIT 

740 - 1/21/11 delivery date good for now 

747 - Vertley testing, may take 2 more weeks 

749 - Will be redelivered once GMAO files with new names are available 

755 - Released to the ASDC on 11/24/10 

756 - 12/3/10 new target release date 

763 - In CM queue 

765 - 1/14/11 delivery date good for now 

770 - 12/10/10 new target release date 

782 - 1.5 more weeks to complete SIT 

787 - Low on priorities list 

795 - Will be redelivered next week 

796 - Sharon is testing 

797 - Sharon is testing 

798 - Promoted 

799 - Low priority for testing 

801 - Low priority for testing 

803 - Waiting to promote 
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804 - Waiting on TSI output data 

807 - Might be promoted next week 

809 - 12/10/10 earliest possible delivery date 

811 - Running a PR in production before promotion 

812 - Promoted 

813 - Promoted 

814 - Will be delivered 4 weeks after Clouds 

816 - Waiting for promotion 

817 - Will deliver soon 

818 - Waiting for promotion 

 

Promotion Priorities List  

Jonathan will distribute an update. 

Additions to the Posted Reports 

Toolkit 

When ASDC recompiled HDF4, a patch was left out.  This has been corrected on AMI & AMI-P. 

 

Instrument 

Have encountered files disappearing and deleted files reappearing on AMI-P.  Will submit a 

consolidated help ticket. 

 

ERBE-like 

Norman wants spectral responses redone for Aqua Edition3.  HDF EOS might need to be 

recompiled for ERBE-like snow and ice to work on AMI-P6 platforms similar to the patch 

for HDF4; meeting Monday to discuss solution. 

 

Clouds 

Clouds science folks are providing input to CPOB meeting status report. 

 

Convolution 

Gathering information on Edition3 CERESlib. 

 

SD_Archive directory – pointers to archive have disappeared, files still exist, just no pointer to 

them.  SAs trying to recover files. 

 

Inversion 

Ran data for Edition4 CRS 

 

SARB 

Ran FSWs for Tom Charlock 

 

G5 names are changing 

 

TISA Gridding & GGEO 

No updates 
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TISA Averaging 

Working on SRBAVG-lite 

 

Files still randomly missing data, might have something to do with times being wrong on blades 

 

Helping Hazari with ISCCP-D2like-Flux 

 

PRs 

No updates 

 

CM 

No updates 

 

CERESlib & Perl_Lib 

No updates 

 

NPP 

No updates 

 

Meeting ended at 2:00 p. m. 


